Author's response to reviews

Title: Human Physiologically Based Pharmacokinetic Model for Propofol

Authors:

David G Levitt (levit001@umn.edu)
Thomas W Schnider (thomas.schnider@kssg.ch)

Version: 6 Date: 24 March 2005

Author's response to reviews:

In this revision I have made all the formatting changes that were requested in the letter of acceptance. This includes changes in title, author affiliations, and the cropping of the figures. In addition, a number of other minor typographical errors were corrected.